Author's response to reviews

Title: Lipid-lowering pharmacotherapy and socioeconomic status: Atherosclerosis Risk in Communities (ARIC) surveillance study

Authors:

Joseph P Kitzmiller (joseph.kitzmiller@osumc.edu)
Randi E Foraker (Foraker.17@osu.edu)
Kathy M Rose (Kathy_Rose@sra.com)

Version: 2 Date: 7 December 2012

Author's response to reviews:

December 7, 2012

Dear Jigisha Patel, M.R.C.P., Ph.D., Medical Editor BMC

The revised version contains the following changes:

1) We added information regarding IRB reviews and the Declaration of Helsinki to the Methods section.

2) We filled-in the missing information in the Acknowledgements section.

Thank you,

Joseph P Kitzmiller M.D., Ph.D., F.C.P., C.P.I.
Departments of Pharmacology and Biomedical Engineering
The Ohio State University Medical Center
joseph.kitzmiller@osumc.edu
(614) 292-8438
(614) 813-5449